Multinational Firms and Business Cycle Transmission

Dominik Menno^{*}

This version: December 2, 2013

Abstract

This paper studies the effect of foreign direct investment (FDI) on the transmission of international business cycles. In a sample of 12 OECD countries between 1995 and 2006, I find that increases in bilateral FDI linkages are associated with more synchronized investment cycles. I also find that the relation between FDI integration and synchronization of gross domestic product (GDP) is - yet positive - statistically insignificant. I then study a model of international business cycles with an essential role for FDI and shocks to multinational activity. In the model, more FDI openness unambiguously increases investment synchronization while the effect on GDP synchronization is ambivalent. Due to mismeasurement of intangible capital in national accounts, the actual elasticity of output synchronization with respect to FDI integration is underestimated.

Keywords: FDI integration, Co-movement, Investment synchronization, Multinationals, Technology capital, International business cycles

JEL Classification Numbers: E32, F15, F21, F23, F44

^{*}European University Institute, Florence, and RWTH Aachen University. Address: School of Business and Economics, RWTH Aachen University, Templergraben 64, 52064 Aachen, Germany. email: dmenno@eui.eu

1 Introduction

A central question in international macroeconomics is how financial integration affects the international transmission of shocks. The literature on international business cycles has extensively analyzed the effects of financial assets, such as bonds or firm equity, on the long-run average implications of financial integration on international business cycles.¹ In terms of empirics, the literature typically focuses on the link between financial integration and output co-movement and is silent about investment co-movement.² This seems somewhat surprising because one robust prediction of the standard international real business cycle model is the strong negative co-movement of investment when financial markets become more integrated.³ In terms of theory, the literature does not distinguish foreign direct investment (FDI) from other financial assets in regard of its consequences for international business cycle co-movement. Yet, FDI is special in that it involves technology flows within boundaries of multinational firms.⁴ In this paper I attempt to make some progress along both the empirical and the theoretical dimension.

Empirical contribution. The main empirical contribution is to document that increases in bilateral FDI linkages are associated with more investment synchronization. This is a potential channel through which multinationals affect international business cycles that has been overlooked so far. In the benchmark regressions, I use a panel data-set of bilateral FDI linkages and data on GDP and investment synchronization for 12 OECD countries over the period 1995 - 2006. I explicitly exclude the recent crisis period because I do not want to have the results to be driven by a few observations.⁵

I also document that the link between FDI linkages and output synchronization is statistically indistinguishable from zero. This complements earlier work, as previous studies

¹Previous work includes Baxter and Crucini (1995), Kehoe and Perri (2002), Heathcote and Perri (2002). For a theoretical argument that the causality is reverse, see Heathcote and Perri (2004).

²Previous work includes Baxter and Kouparitsas (2005), Imbs (2006), and Hsu, Wu, and Yau (2011).

³See for example Backus, Kehoe, and Kydland (1992), Baxter and Crucini (1995), Kehoe and Perri (2002).

 $^{^4\}mathrm{See}$ McGrattan and Prescott (2009), McGrattan and Prescott (2010), Ramondo and Rappoport (2010) and references therein.

⁵In fact, business cycle correlations jumped up since the onset of the financial crisis in 2007, see Perri and Quadrini (2011).

are inconclusive on this issue. The literature tipically finds a positive effect of financial integration on GDP synchronization.⁶ On the other hand, a recent strand of literature on banking integration suggests that the link between financial integration and business cycle co-movement is negative in normal times and positive in times of financial crisis.⁷

Theoretical contribution. From a theoretical perspective, my contribution is to build a dynamic stochastic equilibrium model of foreign direct investment. For this purpose, I embed technology capital and multinational production into a stochastic two-country real business cycle environment. The first objective of the model is to illustrate a concrete mechanism through which exogenous changes in FDI openness affect business cycle synchronization, in particular investment synchronization, and to study how this mechanism works both under shocks to country-specific aggregate productivity and shocks to multinational activity. The second purpose of the model is to conduct counter-factual analyses in order to shed light on what is driving the weak link between FDI integration and GDP synchronization as found in the data. Measured GDP is distorted because intangible investments by multinationals are expensed. Due to this mismeasurement, I show that the actual elasticity of output co-movement with respect to FDI openness is significantly higher than suggested by measured GDP.

The theory embeds features from the models studied in McGrattan and Prescott (2009) and Ramondo and Rappoport (2010) into a dynamic stochastic real business cycle framework. As in McGrattan and Prescott (2009), multinationals accumulate technology capital. This type of capital can be used simultaneously in different plants located both at home and abroad. Similar to Ramondo and Rappoport (2010), multinationals' productivity is subject to stochastic shocks and these shocks apply to all production units the multinational operates, both within and across country borders. By allowing for multinational-specific shocks, the multinationals itself act as a source of business cycles volatility - on top of affecting the propagation of country-specific shocks that originate in other sectors of the economy. In what follows, I consider different model versions (with

⁶See, for example, Imbs (2004) and Hsu, Wu, and Yau (2011).

⁷See Kalemli-Ozcan, Papaioannou, and Peydró (2013) and Kalemli-Ozcan, Papaioannou, and Perri (2013).

and without multinational-specific shocks) that help to disentangle the role played by each model ingredient for the link between FDI openness and business cycle co-movement.

The model is also related to a growing strand of literature that stresses the importance of intangible capital for economic outcomes, such as asset prices (Eisfeldt and Papanikolaou, 2013), managerial compensation (Lustig, Syverson, and Van Nieuwerburgh, 2011), or the life-cycle of firms (Atkeson and Kehoe, 2005).⁸ More closely related, Johri, Letendre, and Luo (2011) study the role of organizational capital for international investment co-movement; their model, however, abstracts from foreign direct investment and is therefore not suited to study the transition to FDI openness.

In terms of research methodology and presentation of the results, I follow in large parts Kalemli-Ozcan, Papaioannou, and Perri (2013). Their research question, yet related, differs from mine. Their focus is on the relation between international banking linkages and output synchronization and on the role of credit shocks for international co-movement. I view my results as complementary to theirs because both deepen our understanding of the effects of financial integration on the international transmission of business cycles.

The paper is organized as follows. Section 2 reports the empirical methodology and the empirical results. Section 3 introduces the theoretical framework. Section 4 presents the quantitative results. Section 5 concludes.

2 Empirical results

2.1 Empirical specification

The empirical model is given by

$$synch_{i,j,t}^{a} = \theta_t + \gamma FDI_{i,j,t-1} + z'_{i,j,t}\beta + c_{i,j} + u_{i,j,t} \quad \text{for} \quad a = GDP, I.$$

⁸These studies focus on organizational capital only whereas I adopt the broader definition of technology capital by McGrattan and Prescott (2009) that includes Brands, R&D, and organizational capital.

synch^a_{i,j,t} is a time-varying bilateral measure reflecting the synchronization for growth in gross domestic product (a = GDP) and investment growth (a = I), respectively, between countries *i* and *j* in period *t*. One period in the regression setup is one year. The variable $FDI_{i,j,t-1}$ measures bilateral cross-border FDI positions between country *i* and *j* in the previous period (year) and $c_{i,j}$ is a country-pair specific unobserved heterogeneity that captures all time-invariant bilateral factors that affect both FDI integration as well as business cycle and investment synchronization.⁹ I also include time dummies (θ_t) to account for shocks common to all countries. Following Kalemli-Ozcan, Papaioannou, and Perri (2013), in order to separate the relative importance of global and country-specific shocks, I also report results for specifications where only country-specific time trends (g_i and g_j) and where both aggregate and country-specific time trends are included:

$$synch_{i,j,t}^{a} = \theta_t + (g_i + g_j) \cdot t + \gamma FDI_{i,j,t-1} + z'_{i,j,t}\beta + c_{i,j} + u_{i,j,t} \quad \text{for} \quad a = GDP, I.$$

The vector $z'_{i,j,t}$ contains measure for trade linkages in the previous period and the lagged level of income and population.¹⁰

Following Kalemli-Ozcan, Papaioannou, and Perri (2013), I measure business cycle synchronization $(synch_{i,j,t}^{GDP})$ by the negative absolute distance in output growth rates between country *i* and country *j* in quarter *t*:

$$synch_{i,j,t}^{GDP} \equiv -\left| \left(\ln GDP_{i,t} - \ln GDP_{i,t-1} \right) - \left(\ln GDP_{j,t} - \ln GDP_{j,t-1} \right) \right|.$$
(1)

Analogously, cross-country investment synchronization in quarter t is defined as

$$synch_{i,j,t}^{I} \equiv -\left|\left(\ln I_{i,t} - \ln I_{i,t-1}\right) - \left(\ln I_{j,t} - \ln I_{j,t-1}\right)\right|.$$
(2)

The yearly estimates for aggregate and investment synchronization are obtained by av-

⁹Other studies have stressed the importance of country-pair fixed effects, see Kalemli-Ozcan, Papaioannou, and Peydró (2013) and Kalemli-Ozcan, Papaioannou, and Perri (2013).

¹⁰I experimented including other correlates usually used in the business cycle co-movement literature (e.g. Frankel and Rose, 1998; Kose and Yi, 2006; Baxter and Kouparitsas, 2005) such as geography, common language, etc; I find that only lagged GDP and population enters some of the here presented specifications with a statistically significant coefficient. For GDP synchronization, this was also found by Kalemli-Ozcan, Papaioannou, and Perri (2013).

Table 1: Descriptive statistics

			1						
	N	Mean	Sd	Min	Max	p25	p50	p75	p95
Pairwise corr. of hp-filtered GDP	636	0.407	0.401	-0.811	0.952	0.132	0.494	0.751	0.877
Pairwise corr. of hp-filtered Investm.	636	0.288	0.396	-0.848	0.969	0.00720	0.315	0.586	0.879
Synch. of GDP	636	-2.355	1.285	-8.472	-0.187	-2.944	-2.055	-1.425	-0.872
Synch. of Investm.	636	-8.960	5.515	-39.22	-0.378	-11.16	-7.764	-5.236	-2.724
FDI/GDP	636	5.458	13.54	0.0466	118.5	0.462	1.722	4.282	22.76
FDI/total FDI	636	7.713	14.31	0.0762	127.5	1.109	2.978	8.174	30.66
Trade/GDP	636	1.622	1.639	0.0835	7.738	0.484	0.885	2.238	5.346
Trade/total trade	636	3.075	3.503	0.126	20.83	0.933	1.746	4.103	8.763

Notes: The pairwise correlations of GDP and investment are the correlation of hp-filtered real GDP and real gross fixed capital formation, respectively, estimated using 20 quarter-rolling-windows. The GDP (investment) synchronization index is the one defined in equation (1) (equation (2)), the yearly average of the quarter by quarter measure. The synchronization indices are in percent (annualized); FDI and trade ratios are defined in equations (3) and (4), the unit is percent. For a data description see appendix A.

eraging over the quarterly synchronization measures. As a robustness check, appendix B reports estimates where the dependent variables are cross-country correlations of hpfiltered investment and GDP, respectively.

I measure cross-border FDI linkages in two ways. First, I use the sum of bilateral asset and liabilities between countries i and j over the sum of the two countries' GDP in each vear:¹¹

$$\left(\frac{FDI}{GDP}\right)_{i,j,t} \equiv \frac{FDIA_{i,j,t} + FDIL_{i,j,t} + FDIA_{j,i,t} + FDIL_{j,i,t}}{GDP_{i,t} + GDP_{j,t}}.$$
(3)

Second, I use bilateral FDI assets and liabilities divided by the sum of total FDI assets and liabilities of the two countries:¹²

$$\left(\frac{FDI}{TotFDI}\right)_{i,j,t} \equiv \frac{FDIA_{i,j,t} + FDIL_{i,j,t} + FDIA_{j,i,t} + FDIL_{j,i,t}}{FDIA_{i,t} + FDIL_{i,t} + FDIA_{j,t} + FDIL_{j,t}}.$$
(4)

The sample for the empirical analysis consists of 53 country pairs for the years between 1995 and 2006, including the G7 countries. For a data description see Appendix A. The appendix also lists the names of the country pairs that are in the balanced sample.¹³ Table 1 reports the descriptive statistics for the relevant variables in the sample.

¹¹See Kalemli-Ozcan, Papaioannou, and Perri (2013).

¹²I also normalized bilateral FDI positions using total foreign assets and liabilities, finding similar results.

¹³The reason for using the restricted sample is data availability. For these country pairs there are no missing values for bilateral FDI positions and we have a balanced sample. Using the full (unbalanced) sample with 18 countries from 1985 to 2006 does not alter the main conclusions. Appendix B reports additional estimation results and robustness checks.

2.2 FDI linkages and investment synchronization

Dependent Variable: Investment growth synchronization (annualized)								
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
FDI/GDP	4.662^{***} (3.01)	4.115^{**} (2.54)	4.106^{**} (2.48)	4.078^{**} (2.50)				
$\mathrm{Trade}/\mathrm{GDP}$				$5.986 \\ (1.03)$				
FDI/Total FDI					3.922^{**} (2.46)	3.739^{**} (2.44)	3.818^{**} (2.48)	3.808^{**} (2.50)
Trade/Total trade								$5.874 \\ (1.19)$
Country-pair fixed Time fixed Country trends R-squared (within) Observations	Yes No Yes 0.036 530	Yes Yes No 0.071 530	Yes Yes V.094 530	Yes Yes Yes 0.099 530	Yes No Yes 0.027 530	Yes Yes No 0.068 530	Yes Yes Yes 0.092 530	Yes Yes Yes 0.098 530

Table 2: Bilateral FDI Linkages and Investment synchronization

Notes: The table reports panel (country-pair) fixed-effect coefficients estimated over the period 1995 to 2006 using 53 country pairs for which bilateral FDI positions were available for all years. A list of countries used can be found table 8 in appendix A. The dependent variable is minus one times the absolute value of the difference in quarterly growth rate of aggregate investment between country i and j in year t (the yearly estimate is obtained by averaging over the respective four quarterly estimates). In columns (1) - (4) FDI integration is measured by the log of the share of the stock of bilateral Foreign Direct Investment Assets and Liabilities between countries i and j in the previous year relative to the sum of the two countries' GDP in the previous year (denoted FDI/GDP). In columns (5) - (8) FDI integration is measured by the log of the share of the stock of bilateral Foreign Direct Investment Assets and Liabilities between countries i and jin the previous year relative to the sum of the two countries' total FDI Assets and FDI Liabilities in the entire world in the previous year (denoted FDI/Total FDI). All specifications also include the log of the two countries' GDP and the log of the product of the two countries' population in the previous year. The specification in (4) includes the log of the share of bilateral export and import flows between countries i and j in the previous year relative to the sum of the two countries' GDP in the previous year (Trade/GDP). The specification (8) includes the log of the share of bilateral export and import flows between countries i and j in the previous year relative to the sum of the two countries' total exports and imports in the previous year (Trade/Total Trade). The specifications in columns (1) and (5) include country-specific linear time-trends. The specifications in columns (2) and (6) include time fixed-effects. The specifications in columns (3), (4), (7), and (8) include time fixed-effects and country-specific linear time-trends. Standard errors adjusted for panel (country-pair) specific auto-correlation and heteroskedasticity and corresponding t-statistics are reported below. * denotes significance at the 90% confidence level, ** denotes significance at the 95% confidence level, *** denotes significance at the 99% confidence level. For a detailed data description see appendix A.

This section reports the findings on the relation between FDI integration and international investment synchronization. Table 2 reports the benchmark estimates on the relation between FDI integration and investment synchronization during the period 1995- 2006. The specification in column (1) controls for country-pair fixed effects and country specific time trends. The coefficient is positive and statistically different from zero. That is, conditional on country specific shocks, within country-pair increases in FDI integration are associated with more synchronized investment. In column (2), I include time fixed-effects to account for common global shocks, while column (3) reports results with time fixed-effects and country-specific time trends. In both specifications, the coefficient on FDI integration is positive and statistically different from zero. In column (4), I control

for bilateral trade linkages.¹⁴ The coefficient on goods trade is positive and similar in magnitude as the coefficient on FDI integration. Yet we cannot reject the Null of a zero coefficient. Most importantly, when controlling for goods trade does not affect the coefficient on FDI integration.¹⁵

To get a sense for the magnitudes, note that FDI linkages are expressed in logs and investment synchronization is in percentage points, hence the coefficients reflect semielasticities. The coefficient in column (3) implies that a doubling in bilateral integration is associated with an average increase in investment synchronization of 4.1 percentage points. Given the median investment synchronization (-7.8 percent) these are significant effects.

Columns (5) to (8) report the results using the alternative FDI integration index as defined in equation (4). The results are similar to the ones presented in columns (1) to (4). More FDI linkages are associated with higher investment synchronization; the point estimates are somewhat lower than the ones in specifications (1) to (4).

The estimated coefficients are robust to a number of robustness checks. In particular, using the full unbalanced sample for all available country pairs from 1985 to 2006 as well as restricting the sample to the G7 country-pairs only, does not change the main result; higher FDI linkages are associated with more investment synchronization. This result holds also in regressions with cross-country correlations of hp-filtered investment as a dependent variable. Appendix B contains more details on these robustness checks.

2.3 FDI Linkages and GDP synchronization

In this section, I present the results of the benchmark estimations for the relation between integration and business cycle correlation. Table 3 reports the benchmark estimates on the effect of FDI linkages on GDP synchronization in the period 1995-2006. In

¹⁴Similar to FDI linkages, bilateral trade is defined as the log of the sum of bilateral trade flows divided by the sum of the countries' GDP.

¹⁵Earlier work (e.g. Frankel and Rose (1998) or Kose and Yi (2006)) showed the importance of trade for aggregate business cycle co-movement. The positive point estimates suggest the existence of some complementarity between FDI and trade. Yet, the trade linkages are only moving slowly over time, so there might be too little within-country correlation to pick up significant effects.

Dependent Variable: GDP growth synchronization (annualized)								
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
FDI/GDP	0.587^{**} (2.34)	$\begin{array}{c} 0.145 \ (0.52) \end{array}$	$\begin{array}{c} 0.133 \ (0.46) \end{array}$	$0.129 \\ (0.45)$				
$\mathrm{Trade}/\mathrm{GDP}$				$\begin{array}{c} 0.945 \\ (1.46) \end{array}$				
$\mathrm{FDI}/\mathrm{Total}\ \mathrm{FDI}\ \mathrm{links}$					$\begin{array}{c} 0.392 \\ (1.58) \end{array}$	$0.184 \\ (0.70)$	$0.178 \\ (0.65)$	$\begin{array}{c} 0.177 \ (0.65) \end{array}$
Trade/Total trade links								1.001^{*} (1.68)
Country-pair fixed	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Time fixed	No	Yes	Yes	Yes	No	Yes	Yes	Yes
Country trends	Yes	No	Yes	Yes	Yes	No	Yes	Yes
R-squared (within)	0.024	0.202	0.221	0.224	0.019	0.202	0.222	0.225
Observations	530	530	530	530	530	530	530	530

Table 3: Bilateral FDI Linkages and GDP synchronization

Notes: The table reports panel (country-pair) fixed-effect coefficients estimated over the period 1995 to 2006 using a balanced panel for 53 country-pairs for which bilateral FDI positions are available in every year. A list of included countries-pairs can be found in appendix A. The dependent variable is minus one times the absolute value of the difference in quarterly growth rates of GDP between country i and j in year t (the yearly estimate is obtained by averaging over the respective four quarterly estimates). In columns (1) - (4) FDI integration is measured by the log of the share of the stock of bilateral Foreign Direct Investment Assets and Liabilities between countries i and j in the previous year relative to the sum of the two countries' GDP in the previous year (denoted FDI/GDP). In columns (5) - (8) FDI integration is measured by the log of the share of the stock of bilateral Foreign Direct Investment Assets and Liabilities between countries i and j in the previous year relative to the sum of the two countries' total FDI Assets and FDI Liabilities in the entire world in the previous year (denoted FDI/Total FDI). All specifications also include the log of the two countries' GDP and the log of the product of the two countries' population in the previous year. The specification in (4) includes the log of the share of bilateral export and import flows between countries i and j in the previous year relative to the sum of the two countries' GDP in the previous year (Trade/GDP). The specification (8) includes the log of the share of bilateral export and import flows between countries i and j in the previous year relative to the sum of the two countries' total exports and imports in the previous year (Trade/Total Trade). The specifications in columns (1) and (5) include country-specific linear time-trends. The specifications in columns (2) and (6) include time fixed-effects. The specifications in columns (3), (4), (7), and (8) include time fixed-effects and country-specific linear time-trends. Standard errors adjusted for panel (country-pair) specific auto-correlation and heteroskedasticity and corresponding t-statistics are reported below. * denotes significance at the 90% confidence level, ** denotes significance at the 95% confidence level, *** denotes significance at the 99% confidence level. For a detailed data description see appendix A.

column (1), controlling for country-pair fixed effects and country specific time trends, the coefficient is positive and statistically different from zero. That is, conditional on country specific shocks, within country-pair increases in FDI integration are associated with more synchronized investment. However, this result is not robust to the inclusion of an aggregate time trend. In column (2), I include time fixed-effects to account for common global shocks, while column (3) reports results with both time fixed-effects and countryspecific time trends. In both specifications, the coefficient on FDI integration remains positive but is statistically indistinguishable from zero. Specification (4) controls for bilateral trade linkages. The coefficient on goods trade is positive and bigger in size than the coefficient on FDI integration. Yet we cannot reject the Null of a zero coefficient.¹⁶

¹⁶A reason for this finding could also be reverse causality: less correlated country pairs engage in more FDI. In this case, the presented coefficients presented here are lower bounds as this argument describes

From the table also emerges that controlling for goods trade does not affect the finding of a quantitative small and statistical insignificant link between FDI integration and GDP synchronization.

Columns (5) to (8) report the results using the alternative FDI integration index as defined in equation (4). The results are similar to the ones in columns (1) to (4), except that in specification (8) the coefficient on trade linkages becomes statistically significant at the 10 percent level. The results regarding FDI integration and GDP synchronization remain unchanged: there is no statistical significant link between FDI linkages and GDP synchronization.

3 A model of international business cycles with foreign direct investment

In this section, I develop a model of international business cycles where multinationals accumulate technology capital and engage in FDI. Technology capital is firm-specific and can be simultaneously used in multiple plants in locations at home and abroad.¹⁷ The plants operated by multinationals thus produce all with the same technology capital. There are two types of shocks causing economic fluctuations: a standard country-specific productivity shock and a shock that is multinational-specific, affecting the efficiency of the existing technology capital. This multinational-specific shock therefore affects both the returns on domestic and on foreign investment.

The model serves three purposes. The first is to precisely lay out a causal link between FDI openness and international investment synchronization. The empirical section documents a relationship between the two, but does not speak about the underlying mechanism and the direction of causation. I will use the model to derive quantitative results that show how the empirical findings are indeed consistent with the hypothesis that FDI openness has significant effects on investment synchronization. The second purpose of

a downward bias for the un-instrumented estimates. For the theoretical argument, see (Heathcote and Perri, 2004).

 $^{^{17}}$ For the concept of locations in this context refer to McGrattan and Prescott (2009).

the model is to shed light on the weak link between FDI openness and GDP synchronization, as documented in the empirical section. For this purpose, I measure GDP in the model in the same way as in national accounts data where investments in intangible capital are expensed. With the quantitative results of the model, I show that FDI openness has indeed weak effects on business cycle synchronization when using measured GDP as a proxy for aggregate activity. Third, and relatedly, I use the model to conduct a counter-factual analysis to show that the relation between FDI openness and business cycle synchronization is significantly stronger when aggregate output is measured correctly.

The framework combines earlier work from Ramondo and Rappoport (2010) and Mc-Grattan and Prescott (2009) to incorporate multinational production into an international business cycles set-up. The main innovation is that I consider both a stochastic environment and allow for an explicit role for FDI; as such, the set-up is well-suited to analyze the effects of cross-border FDI integration on investment synchronization. As will be shown below, a key ingredient of the model is the accumulation of technology capital.

3.1 The economy

I consider a two-countries, two-sectors, two-goods world. In each country (foreign variables are denoted by an asterisk), there are households of equal mass normalized to unity that consume a tradable final consumption good and supply labor to firms. Firms in the final good sector buy intermediate inputs from intermediate good firms, hire labor, accumulate physical capital and pay wages and dividends to domestic households. Physical capital and labor are not mobile across countries but across sectors. The intermediate good is not tradable across countries and producers in this sector buy differentiated goods from domestic and foreign firms, labelled multinationals. Multinationals can accumulate physical capital in both countries and set up production units both at home and abroad through which they serve the foreign intermediate goods market. Multinationals pay dividends to their owners, domestic multinationals are entirely owned by domestic households and foreign multinationals are entirely owned by foreign households.¹⁸ In addition to physical capital, multinationals accumulate technology capital. Technology capital is firm-specific and can be used in multiple locations in both countries at the same time. For international financial markets, I consider two scenarios: one in which international financial markets are complete in the sense that households have access to a full set of state-contingent securities that can be traded internationally. The other scenario is one in which households cannot trade any international assets and just receive labor income and dividends from domestic firms and multinationals.

Time and uncertainty. Time is discrete and denoted by t = 1, 2, ... In each period t the economy experiences one event $s_t \in S$ where S is a possibly infinite set. I denote by s^t the history of events up to and including date t. The probability at date 0 of any particular history s^t is given by $\pi(s^t)$. For the sake of readability (and with some abuse of notation), I will drop the explicit reference to histories and states most of the time when there is no room for confusion; I will use the subscript t instead to refer both to the time period and histories.

Households. Households supply labor and the total supply of time is normalized to \bar{L} ; households derive utility from consumption of the perishable good C_t and from leisure $\bar{L} - L_t$. Households maximize the expected discounted sum of future period utilities given by

$$E\sum_{t=0}^{\infty}\beta^t U(C_t, \bar{L} - L_t)$$

where E represents expectations across all possible states of the world, C_t denotes consumption, L_t is labor effort, $0 < \beta < 1$ is the discount factor, and the period-by-period utility function is given by $U(C_t, \bar{L} - L_t) = \log(C_t) + \alpha \log(\bar{L} - L_t)$. Given aggregate wages w_t , households receive labor income $w_t L_t$ and dividend payments from domestic tradable good firms d_{Tt} and from multinationals d_{Mt} , respectively.

 $^{^{18}\}mathrm{I}$ exclude that firm shares are traded. Because financial markets are complete this is without loss of generality.

International financial markets. I consider two versions of the model, one with complete international financial markets and one with incomplete financial markets in the sense that households cannot trade any international assets.

1. In the **complete** financial markets scenario, households have available a complete set of Arrow securities. Let $B_t(s^t, s_{t+1})$ be the quantity of bonds purchased by the home households at time t after history s^t that pay one unit of the consumption good in t + 1 if and only if the state of the world economy in t + 1 is equal to s_{t+1} . Let $q_t(s^t, s_{t+1})$ be the price of such a bond. Under complete international financial markets, the budget constraint for the representative household in the home country is

$$C_t + \sum_{s_{t+1}} q_t(s^t, s_{t+1}) B_t(s^t, s_{t+1}) = w_t L_t + d_{Tt} + d_{Mt} + B(s^{t-1}, s_t)$$
(5)

and the budget constraint for foreign households is analogously defined.

2. Under **incomplete** international financial markets, households are not allowed to trade any financial asset across country borders. In this model version, the budget constraint for the representative household in the home country is

$$C_t = w_t L_t + d_{Tt} + d_{Mt},\tag{6}$$

analogous for the foreign households.

Firms in the tradable goods sector. The tradable consumption good is produced under perfect competition with a constant returns to scale technology that combines labor (l_{Tt}) , capital (k_{Tt}) and the composite intermediate good (X_t) . Production in the this sector is subject to stochastic and country-specific productivity shocks a_t and a_t^* . Firms' production function is given by

$$Y_{t} = e^{a_{t}} \left(k_{Tt}^{\theta} l_{Tt}^{1-\theta} \right)^{\nu} X_{t}^{1-\nu}, \tag{7}$$

where $0 < \nu < 1$. Final good firms purchase X_t units of the intermediate good from competitive intermediate good producers at a unit price P_t , where I normalized the price

of the tradable good to one. Firms' dividends are thus given by

$$d_{Tt} = e^{a_t} \left(k_{1t}^{\theta} l_{1t}^{1-\theta} \right)^{\nu} X_t^{1-\nu} - P_t X_t - w_t l_{Tt} - i_{Tt}$$
(8)

where i_{1t} represents investment in physical capital. The capital stock evolves according to

$$k_{Tt+1} = (1-\delta)k_{Tt} + \left[\frac{\chi_1\left(\frac{i_{Tt}}{k_{Tt}}\right)^{1-\psi}}{1-\psi} + \chi_2\right]k_{Tt}$$
(9)

where δ is the depreciation rate, ψ determines the sensitivity of the cost to investment, and the parameters χ_1 and χ_2 are set by imposing steady state targets.¹⁹ I assume that the productivity shocks follow a bivariate auto-regressive process

$$\begin{bmatrix} a_t \\ a_t^* \end{bmatrix} = \Lambda_a \begin{bmatrix} a_{t-1} \\ a_{t-1}^* \end{bmatrix} + \begin{bmatrix} \varepsilon_t^a \\ \varepsilon_t^{a^*} \end{bmatrix}$$
(10)

where Λ_a is a 2×2 matrix and $[\varepsilon_t^a, \varepsilon_t^{a^*}]'$ is a vector of i.i.d. random variables with mean 0, standard deviation σ_a and correlation ρ_{ε}^a . The problem of domestic tradable goods firms is then

$$\max E \sum_{t}^{\infty} Q_t d_{Tt}$$

subject to (8), (9), k_{10} given, where $Q_t = \beta^t U_c(C_t, L_t)$ is the marginal utility of period t consumption of domestic consumers who are the owners of the firm. The problem of tradable good firms in the foreign country is analogous.

Intermediate good producers. Intermediate good producers buy non-tradables produced by multinationals and sell the bundled good Y_{It} at price P_t to final good producers. The index Y_{It} aggregates a continuum of intermediate goods with a constant elasticity of substitution $\frac{1}{1-\eta}$. As discussed in more detail below, I assume that there are only two types of firms in each country: (i) domestic multinationals and (ii) foreign multinationals.

¹⁹This functional form is widely used in the literature, see for example Quadrini and Jermann (2012). The parameters are chosen such that the depreciation rate is equal to δ and that the derivative of capital with respect to investment is equal to one.

Given this assumption, intermediate good producers' output reads as

$$Y_{It} = \left[x_{dt}^{\eta} + x_{ft}^{\eta} \right]^{\frac{1}{\eta}}$$
(11)

where x_{dt} and x_{ft} denote the intermediate good producers' demand for goods produced by domestic and foreign multinationals, respectively. The implied demand functions are given by

$$x_{jt} = (P_t/p_{jt})^{\frac{1}{1-\eta}} \qquad j = d, f.$$
 (12)

The main effect of adding imperfect competition to the model is that it scales up the amount of variable profits in the economy; hence, it scales up the size of the payments owners receive from technology capital, something that does not affect the qualitative implications of the model but is necessary to obtain realistic amounts of FDI when undertaking the quantitative analysis below.

Multinationals. In both countries, there is a large number of locations where production can take place.²⁰ The measure of locations is, without loss of generality, normalized to one. In each location, both domestic and foreign multinationals can set up a plant and operate. The production of a plant owned by a domestic multinational in a given location *i* depends on firm specific productivity z_t , labor services $l_t(i)$ and physical capital $k_t(i)$ and is given by a decreasing returns to scale technology $y_t(i) = e^{z_t}(k_t(i)^{\theta}l_t(i)^{1-\theta})^{1-\phi}$ with $0 < \phi < 1$. While physical capital and labor are both specific to each multinational and plant, technology capital M_t and productivity z_t is specific to each multinational only. The productivity of the foreign multinational is denoted by z_t^* . Technology capital and productivity therefore affect production in all locations, both domestic and foreign, at which the firm operates. A home multinational with M_t units of technology capital, k_{dt} units of domestic physical capital, and l_{dt} units of domestic labor services efficiently allocates physical capital and labor across all M_t domestic plants. Therefore, its total

 $^{^{20} {\}rm The}$ derivation of the multinationals' production technology follows closely McGrattan and Prescott (2009); see also Kapicka (2012).

production in the home country is given by

$$y_{dt} = e^{z_t} M_t^{\phi} \left(k_{dt}^{\theta} l_{dt}^{1-\theta} \right)^{1-\phi}.$$
 (13)

Technology capital can also be used to set up operations in a foreign location. Foreign owned multinationals accumulate domestic physical capital and hire domestic labor services and use their own technology capital. In contrast to domestic firms, the production of a foreign multinational depends on the countries' FDI openness. The degree of openness to FDI for both countries is given by a parameter τ that determines the total average factor productivity of a foreign multinational relative to a domestic multinational.²¹ To illustrate this point, consider a multinational owned by the domestic consumer with given technology capital M_t and productivity z_t . It allocates efficiently its foreign physical capital k_{ft} and foreign labor services l_{ft} to generate total output abroad given by

$$y_{ft} = \tau e^{z_t} M_t^{\phi} \left(k_{ft}^{\theta} l_{ft}^{1-\theta} \right)^{1-\phi}.$$
(14)

Analogously, a foreign owned multinational with M_t^* units of technology capital and productivity z_t^* , k_{ft}^* units of home country's physical capital, and l_{ft}^* units of home country's labor services produces total output in the home country according to

$$y_{ft}^* = \tau e^{z_t^*} (M_t^*)^{\phi} \left((k_{ft}^*)^{\theta} (l_{ft}^*)^{1-\theta} \right)^{1-\phi}.$$
 (15)

The domestic multinationals' total dividends are then given by the proceeds from their domestic and foreign operations, respectively, or

$$d_{Mt} = (p_{dt}y_{dt} - w_t l_{dt} - i_{dt}) + (p_{ft}^* y_{ft} - w_t^* l_{ft} - i_{ft}) - i_{Mt}$$
(16)

where the inverse demand functions p_{dt} , p_{ft}^* of domestic and foreign intermediate good producers defined in (12) are taken as given; i_{dt} and i_{ft} represent investment in domestic and foreign physical capital, respectively, and i_{Mt} represents the multinationals' invest-

 $^{^{21}}$ Here, we impose symmetry across countries and assume that both countries have the same degree of openness.

ment in technology capital. The respective capital stocks evolve according to

$$k_{jt+1} = (1-\delta)k_{jt} + \left[\frac{\chi_1\left(\frac{i_{jt}}{k_{jt}}\right)^{1-\psi}}{1-\psi} + \chi_2\right]k_{jt} \qquad j = d, f$$
(17)

$$M_{t+1} = (1 - \delta_m)M_t + \left[\frac{\chi_1^m \left(\frac{i_{Mt}}{M_t}\right)^{1 - \psi_m}}{1 - \psi_m} + \chi_2^m\right]M_t$$
(18)

where δ_m is the depreciation rate of technology capital, ψ_m determines the sensitivity of the cost to investment in technology capital, and the parameters χ_1^m and χ_2^m are set by imposing steady state targets.²² Note that the parameters for the adjustment costs in physical capital are identical across sectors. To complete the description of the multinationals problem, I assume that the log of domestic and foreign multinationals' productivity evolves according to a bivariate auto-regressive process

$$\begin{bmatrix} z_t \\ z_t^* \end{bmatrix} = \Lambda_z \begin{bmatrix} z_{t-1} \\ z_{t-1}^* \end{bmatrix} + \begin{bmatrix} \varepsilon_t^z \\ \varepsilon_t^{z^*} \end{bmatrix}$$
(19)

where Λ_z is a 2 × 2 matrix and $[\varepsilon_t^z, \varepsilon_t^{z^*}]'$ is a vector of i.i.d. random variables with mean 0, standard deviation σ_z and zero correlation. Multinationals with origins in the home country solve

$$\max E \sum_{t}^{\infty} Q_t d_{Mt}$$

subject to (12),(16), (17), (18), M_0 , k_{d0} , k_{f0} given, where Q_t again is the marginal utility of consumption of the domestic consumers (who are the owners). The problem of foreign owned multinationals is analogous.

3.2 Equilibrium

An equilibrium, for an exogenously given level of FDI openess τ , is a collection of price sequences p_{dt} , p_{dt}^* , p_{ft} , p_{ft}^* , P_t , P_t^* , Q_t , Q_t^* , $q(s^t, s_{t+1}) \forall s_{t+1} \in S$, exogenous shock processes

 $^{^{22}}$ I set the parameters such that the depreciation rate in steady state is equal to δ_m and the derivative of technology capital with respect to investment is equal to one.

 $z_{t}, z_{t}^{*}, a_{t}, a_{t}^{*} \text{ and quantities } C_{t}, L_{t}, i_{Tt}, i_{dt}, i_{ft}, i_{Mt}, l_{Tt}, l_{dt}, l_{ft}, d_{t}, x_{dt}, x_{ft}, y_{dt}, y_{ft}, X_{t}, Y_{It}, Y_{t}, B(s^{t}, s_{t+1}), C_{t}^{*}, L_{t}^{*}, i_{dt}^{*}, i_{ft}^{*}, i_{Mt}^{*}, l_{Tt}^{*}, l_{dt}^{*}, l_{ft}^{*}, d_{t}^{*}, x_{dt}^{*}, x_{ft}^{*}, y_{dt}^{*}, y_{ft}^{*}, X_{t}^{*}, Y_{It}^{*}, Y_{t}^{*}, B^{*}(s^{t}, s_{t+1}) \forall s_{t+1} \in S \text{ such that:}$

- 1. Given prices and shocks, consumers and firms solve their respective problems.
- 2. Labor markets clear, i.e.

$$L_t = l_{dt} + l_{ft}^* + l_{Tt} \quad \text{for all } t.$$
$$L_t^* = l_{dt}^* + l_{ft} + l_{Tt}^* \quad \text{for all } t.$$

3. Intermediate goods markets clear, i.e.

$$\begin{aligned} X_t &= Y_{It} & X_t^* &= Y_{It}^* & \text{for all } t. \\ x_{dt} &= y_{dt} & x_{ft} &= y_{ft}^* & x_{dt}^* &= y_{dt}^* & x_{ft}^* &= y_{ft} & \text{for all } t. \end{aligned}$$

4. Under complete financial markets bond markets clear, i.e.

$$B(s^t, s_{t+1}) + B^*(s^t, s_{t+1}) = 0$$
 for all $t, s_{t+1} \in S$.

5. The tradable goods market clears, i.e.

$$C_t + C_t^* + i_{Tt} + i_{Tt}^* + i_{dt} + i_{ft} + i_{dt}^* + i_{ft}^* + i_{Mt} + i_{Mt}^* = Y_t + Y_t^*$$
 for all t.

3.3 National accounts and measured returns

Because in national accounts investment in technology capital is expensed,²³ measured gross domestic product in the home country is given by

$$GDP_t = Y_t - i_{Mt}.$$
(20)

 $^{^{23}}$ see McGrattan and Prescott (2010).

This means that GDP differs from actual value added Y_t whenever investment in technology capital is different from zero. This also implies that the dynamic properties of GDP - in particular cross country correlations - depend both on output and investment in technology capital.

Gross FDI positions are given by

$$FDIA_t = k_{ft}$$
 $FDIL_t = k_{ft}^*$

and total bilateral FDI linkages are computed in line with the empirical estimates²⁴

$$FDI/GDP = \frac{2(FDIA_t + FDIL_t)}{4(GDP_t + GDP_t^*)}.$$

In terms of measurement, other key variables are the returns on FDI. Returns reported in balance of payment statistics e.g. by the BEA do not coincide with the *actual* returns multinationals receive from foreign direct investment.²⁵ To see this in the present setup, consider the actual return domestic multinationals receive from their subsidiaries abroad

$$r_{ft} = \theta \eta (1 - \phi) \frac{p_{ft}^* y_{ft}}{k_{ft}} - \delta.$$

In the data, measured returns of foreign subsidiaries from the abroad are computed as FDI income (dividends plus reinvested earnings) divided by the tangible capital stock owned by the multinationals. In the notation of my model, measured returns for the domestic multinational from its subsidiaries abroad are given by

$$r_{FDI,t} = \frac{p_{ft}^* y_{ft} - w_t^* l_{ft} - \delta k_{ft}}{k_{ft}} = r_{ft} + (1 - (1 - \phi)\eta) \frac{p_{ft}^* y_{ft}}{k_{ft}}.$$
 (21)

As the returns on technology capital are not taken into account, measured returns differ from the actual returns by the second term in the above expression. In order to calibrate of the multinational-specific shock, I will match the volatility of measured returns published by the BEA, as outlined in more detail in the next sub-section.

 $^{^{24}{\}rm Note}$ that quarterly GDP is annualized because in national accounts, quarterly gross domestic product is reported at annualized levels.

²⁵This was first pointed out by McGrattan and Prescott (2010).

Table 4: Parameter values							
Description	Symbol	Model					
		Complete Markets	Complete markets, both shocks	Incomplete markets, both shocks			
Preferences							
Discount factor	$egin{array}{c} eta \ ar{L} \end{array}$	0.99					
Total time endowment	\bar{L}	3					
Weight of leisure	α	1.602	1.602	0.955			
Technology							
Income share of labor in production	θ	0.31					
Share of intermediate goods in tradables	ν	0.50					
Production share of technology capital	ϕ	0.21					
Elasticity of demand interm. good sector	$\frac{1}{1-\eta}$	10					
Degree of FDI openness	au	0.23	0.23	0.23			
Deprectiation and adjustment costs							
Depreciation physical capital	δ	0.025					
Depreciation technology capital	δ_m	0.0375					
Adjustment cost physical capital	ψ	0.097	0.099	0.138			
Adjustment cost technology capital	ψ_m	0.089	0.095	0.122			
Std. dev. prod. shock	σ_a	0.008	0.007	0.009			
Std. dev. techn. capital efficiency shock	σ_z	0.000	0.009	0.008			
Cross-country correlation prod. shock	$ ho_a$	0.660	0.920	0.570			
Autoregessive coefficients	$\Lambda_a = \Lambda_z$	$\begin{bmatrix} 0.95 & 0 \\ 0 & 0.9 \end{bmatrix}$	5]				

3.4 Calibration

The equilibrium described above does not admit an analytical solution. I therefore derive a numerical solution using standard linearization techniques. For this purpose, I need to assign numerical values to the various parameters. Table 4 shows the parameters used in the calibration for the three model specifications considered: (1) the model with complete financial markets and country-specific productivity shocks only, (2) the model with complete financial markets with both country-specific productivity shocks and mutlinational-specific shocks to the efficiency of technology capital, and (3) incomplete financial markets with both shocks.

The discount factor is set to $\beta = 0.99$ implying an average interest rate of 4 percent. The

share of intermediate inputs in final tradable production ν is set to 0.5, following Alvarez and Lucas (2007).²⁶ For the demand elasticity of substitution in the intermediate sector I follow Atkeson and Kehoe (2005) and set η to 0.9, implying mark-up of 11 percent and an elasticity of substitution of 10. For the depreciation of physical capital, I choose a standard value and set δ equal to 0.025. Regarding the depreciation of technology capital, I use $\delta_m = 0.0375$, implying an annual depreciation rate of 15 percent, that is, the BEA estimate for depreciation of R&D capital.²⁷

As outlined in the following paragraphs, the remaining parameters are chosen in order to match key moments of the data. This includes also the parameters for the stochastic processes because - even in the model with country-specific productivity shocks only (a_t, a_t^*) - there is not a one-to-one mapping between the stochastic processes and the Solow residual obtained from the data because GDP is mismeasured. Table 5 reports the data targets and the model fit.

Utility and production. The weight on leisure in the utility function α is set so that households, on average, work one third of the available time. By normalizing the total time endowment to $\overline{L} = 3$, this implies a long-run target for employment equal to one. For the share of technology capital in multinationals' production (ϕ), I follow McGrattan and Prescott (2009) and match average investment in technology capital over GDP equal to eight percent. The share of capital in production (θ) is then set so that in steady state the labor share is 64 percent. The degree of FDI openness (τ) is set so that the bilateral FDI position in the model - measured as in the data according to equation (3) - is equal to 1.7 percent, that is, the median value of bilateral FDI linkages in the data, shown in table 1 row five, column seven.

 $^{^{26}}$ Even though in their model the only input in production is labor, the model is calibrated in a way that is perfectly consistent to my setup, as they compute the share of effective labor which includes capital.

 $^{^{27}}$ See Eisfeldt and Papanikolaou (2013). Also McGrattan and Prescott (2010) or Kapicka (2012) assume that technology capital depreciates faster than physical capital. For the main results of this paper, this assumption is not crucial. The quantitative implications are affected but not sensitive for reasonable perturbations of this parameter (between 0.01 and 0.15).

	Data	Model		
		Complete Markets	Complete markets, both shocks	Incomplete markets, both shocks
Long-run averages				
Labor income share	64	64	64	64
Investment in technology capital over GDP	8	8.0	8.0	8.0
$\mathrm{FDI}/\mathrm{GDP}$	1.7	1.7	1.7	1.7
Second moments				
Std. dev. of GDP	1.3	1.3	1.3	1.3
Std. dev. of investment in physical capital relative to GDP	3	3.0	3.0	3.0
Std. dev. of investment in technol- ogy capital relative to GDP	3	3.0	3.0	3.0
Cross-country GDP correlation	0.5	0.5	0.5	0.5
Std. dev. FDI returns	1.6	0.9	1.6	1.6

Table 5: Targeted data moments and model fit

Notes: This table presents the target moments used for the calibration. I compare the moments in the data to the averages of the models' stationary distributions obtained by simulating 150000 time periods and dropping the first 50000. The labor income share and returns on FDI are computed from US data, published by the BEA. The remaining data moments are the median values across time and countries for the OECD sample, see table 1. For a detailed description of the data see appendix A. Column two shows the data moments. Column three refers to the model with complete financial markets and country-specific productivity shocks in the tradable sector only. Column four shows the implied moments from the model with complete financial markets and both country-specific productivity shocks to the efficiency of technology capital. Column five shows the moments for the model with incomplete markets and both shocks.

Adjustment costs. Adjustment costs for both physical capital (ψ) and technology capital (ψ_m), respectively, are set so that the hp-filtered investment series of both types of capital are three times as volatile as hp-filtered GDP. For physical capital, this is a standard value and consistent with OECD data. For technology capital, the number requires some discussion. Ouyang (2011) reports that the growth rate of R&D expenditures are 1.6 times as volatile as GDP growth rates; on the other hand, Eisfeldt and Papanikolaou (2013) find that the volatility of the investment rate in organizational capital is 1.5 percent (annual).²⁸ Targeting the latter value would imply in this model that investment in technolgy capital was six times as volatile as GDP. Because I adopt the broader definition of technology capital (marketing expenditures plus organizational capital plus

 $^{^{28}}$ They define the investment rate as aggregate investment divided by the existing capital stock, see Eisfeldt and Papanikolaou (2013) table II.

R&D) and R&D is the largest part in technology capital,²⁹ I choose a value in between and assume that investment in technology capital behaves similar to investment in physical capital in terms of volatility, in line with the findings in Wälde and Woitek (2004) for R&D investment and G7 data. It is worth noticing that the selected adjustment costs on investment in physical capital are quiet low compared to standard business cycle models; the presence of technology capital makes investment in physical capital less volatile.

Exogenous shocks. For the stochastic processes I assume that the transition matrices are the same $\Lambda_a = \Lambda_z$, with a value of 0.95 on the diagonals and zero on the off-diagonals. Multinational-specific productivity shocks are perfectly correlated within countries (all domestic multinationals have the same productivity) and not correlated across countries (foreign multinationals' productivity follows a statistically independent stochastic process). The volatility of productivity in the tradable sectore (σ_a) and its cross-country correlation (ρ_a) are set so that measured GDP in the model matches the standard deviation of GDP in the data (1.3 percent) and the median value of the hp-filtered cross-country GDP correlation in the data (0.5, see table 1), respectively. For the model specifications with multinational-specific shocks (z_t, z_t^*), I choose the volatility σ_z such that the volatility of measured FDI returns as defined in equation (21) matches the reported volatility in the data, equal to 1.6 percent annualized.³⁰

3.5 Impulse responses

This subsection presents the dynamic responses to the two types of shocks in separation. This serves two purposes. First, I show in what respects multinational-specific shocks differ from country-specific productivity shocks, in terms of model dynamics. Second, this allows me to illustrate how the transmission of these two shocks changes as countries open up to FDI. I therefore present both the impulse responses for country pairs with relatively little FDI linkages (i.e. a low τ) and countries with large FDI linkages (i.e. high τ).

 $^{^{29}\}mathrm{See}$ McGrattan and Prescott (2010).

³⁰Note that the estimate in the data comes from the BEA and is the average over US inward and outward FDI income (see data appendix for a description of the data).



Figure 1: Impulse responses to a productivity shock: low versus high bilateral FDI linkages

Notes: The figure shows impulse responses to a one-standard deviation positive productivity shock of one standard deviation in the home country (ε_t^a). All responses are in percentage deviations from the steady state, normalized by the standard deviation of the shock. Solid lines refer to the relatively closed country pair (low τ), dashed lines refer to the relatively open country pair (high τ). Blue lines refer to the home country, red lines to the foreign country. Panel a) plots the impulse responses for value added (Y_t and Y_t^*) as defined in equation (7), panel b) for measured GDP (defined as $Y_t - i_{mt}$), panel c) for investment in physical capital ($i_t = i_{Tt} + i_{dt} + i_{ft}^*$ and $i_t^* = i_{Tt}^* + i_{dt}^* + i_{ft}$), and panel d) for investment in technology capital (i_{Mt} and i_{Mt}^*).

3.5.1 Productivity shocks

Figure 1 shows the impulse responses to a positive productivity shock in the home country's tradable sector for value added, measured GDP, and investments in physical and technology capital. Blue lines refer to the domestic country, the red lines to the foreign country. Solid lines refer to a country pair that is relatively closed to FDI, dashed lines refer to a relatively open country pair in terms of FDI.³¹ Three observations emerge.

First, both actual value added (panel a) and GDP (panel b) increase in the home country while both show relatively little response in the foreign country. At the same time, investment in physical capital (panel c) and investment in technology capital (panel d) increase in the home country and fall in the foreign country; notably, the on-impact increase of investments in the home country exceeds the fall in the foreign country by a factor four.

Second, technology capital behaves differently to a domestic productivity shock when countries are relatively more open to FDI (dashed lines); while the on-impact increase in technology capital investment in the home country decreases from four to two and a half percent, the response of foreign investment in technology capital switches sign and actually turns positive. Hence, when countries are relatively more open to FDI, a positive productivity shock in the home country benefits both domestic and foreign multinationals, whose dividends increase. Because the shocks are persistent, the expected returns on technology capital increase. Within a firm, in turn, all returns are equalized, therefore the returns both on foreign and domestic investment (from the firms' perspective) increase. As a consequence, foreign multinationals invest not only more in technology capital, but also more in physical capital both at home and abroad. On the other hand, because part of the additional returns of the increased domestic productivity benefits foreign firms, domestic firms increase their investments by a smaller amount than it would be the case when countries are relatively closed to FDI. The previous discussion is reflected by the responses of countries' aggregate investment in physical capital shown by the dashed lines in panel c). The responses shift move closer together, meaning that domestic and foreign investment become less negatively correlated.

Third, actual value added and measured GDP - shown in panels a) and b) - do qualitatively not respond differently when varying the countries' openness to FDI (compare the solid versus dashed lines). The only notable difference is the on-impact response in measured GDP: because investment in technology capital responds less positive in the

³¹For space considerations, the shock process itself is not shown. I consider a one-standard deviation shock in period 1; all plotted responses are normalized by the standard deviation of the shock.



Figure 2: Impulse responses to a positive shock to multinational activity: low versus high bilateral FDI linkages

Notes: The figure shows impulse responses to a one-standard deviation positive shock to the efficiency of domestic multinationals (ε_t^z). All responses are in percentage deviations from the steady state and normalized by the standard deviation of the shock. Solid lines refer to the relatively closed country pair (low τ), the dashed lines refer to the relatively open country pair (high τ). Blue lines refer to the home country, red lines to the foreign country. Panel a) plots the impulse responses for value added (Y_t and Y_t^*) as defined in equation (7), panel b) for measured GDP (defined as $Y_t - i_{mt}$), panel c) for investment in physical capital ($i_t = i_{Tt} + i_{dt} + i_{ft}^*$ and $i_t^* = i_{Tt}^* + i_{dt}^* + i_{ft}$), and panel d) for investment in technology capital (i_{Mt} and i_{Mt}^*).

home country compared to the closed case, GDP increases by more on impact (and the reverse for the foreign country). The transition dynamics however are largely unaffected, suggesting that GDP correlations become somewhat more negative with increasing FDI integration.

3.5.2 Shocks to multinational activity.

Figure 2 shows the impulse responses to a positive shock to domestically owned multinationals' productivity (ε_t^z) for value added, measured GDP, and investments in physical and technology capital, respectively. Blue lines refer to the domestic country, red lines to the foreign country. The solid lines refer to a country pair that is relatively closed to FDI, dashed lines refer to a relatively open country pair in terms of FDI.³²

For relatively closed countries (solid lines) multinational-specific shocks work very much like standard productivity shocks, moving all quantities in a similar fashion as described in the previous sub-section. The reason is that in this case the share of foreign firms is too small to affect aggregate quantities.

On the other hand, the picture changes when countries are relatively open to FDI (dashed lines). First, consider the responses of investment in technology capital shown in panel d). The dashed blue lines lies above the solid blue line, meaning that domestic multinationals increase their investment in technology capital by more than in the closed economy case. The reason is that domestic multinationals gain from investments in technology capital because of the relatively higher returns from abroad. Similarly, the red dashed line lies below the red solid line meaning that foreign multinationals investments in technology capital decrease by more than compared to the case with low FDI linkages.

This, in turn, affects the profitability of investment in physical capital, so it is rational for the multinationals to invest more at home and abroad. This is reflected by the increase in aggregate investment both at home and abroad, as shown by the dashed lines in panel c). Note that this also implies that the increase in investment in physical capital in the home country is below the solid blue line in panel c) because part of the resources get redirected to the foreign country. As a consequence, international investment becomes more synchronized when countries are relatively open to FDI. Hence, multinational-specific shocks exacerbate the effect of technology on the investment comovement.

 $^{^{32}}$ For space considerations, the shock process itself is not shown. I consider a one-standard deviation shock in period 1; all plotted responses are normalized by the standard deviation of the shock.

	GDP	Consump- tion	Invest- ment	Employ- ment	Net Exports
Volatilities		relative to	GDP volat	ility	
Complete markets, prod. shock only	1.30	0.49	3.00	0.52	0.27
Complete markets, both shocks	1.30	0.49	3.00	0.52	0.27
Incomplete markets, both shocks	1.30	0.76	3.00	0.26	0.11
Correlation with GDP					
Complete markets, prod. shock only		0.85	0.97	0.93	0.38
Complete markets, both shocks		0.86	0.97	0.92	0.36
Incomplete markets, both shocks		1.00	0.99	0.97	-0.90
International Correlation					
Complete markets, prod. shock only	0.50	1.00	0.29	-0.14	
Complete markets, both shocks	0.50	1.00	0.28	-0.17	
Incomplete markets, both shocks	0.50	0.50	0.50	0.49	

Table 6: Business cycle statistics

Finally, consider the responses of value added in panel a) and GDP in panel b) in the high FDI linkage scenario (dashed lines). The first notable observation is that value added in both countries increase in response to the shock. The reason is that part of the resources are shifted to the foreign country in form of additional investment of home owned multinationals leading to more output abroad. This is also reflected by the less positive response of value added in the home country (the dashed blue line lies below the solid blue line). Measured GDP - as shown by the dashed lines in panel b) - reflects not only the movements of value added but also the responses in investment in technology capital. Because domestic investment in technology capital responds relatively more when FDI linkages are high, the impulse response of domestic GDP shifts towards South-East such that it even crosses the response of foreign GDP. That means that - on impact - measured GDP abroad increases by more than domestic GDP. In the subsequent transition then, domestic GDP rises, while foreign GDP falls. This suggests that measured GDP might get even more negatively correlated when FDI openness increase.

4 Quantitative Results

4.1 Business cycle properties

In this section, I use the model to assess the business cycle implications of technology capital and shocks to multinational activity. For this purpose, I compare the long-run business cycle statistics for the three model specifications. Table 6 reports the results of the quantitative exercise. The rows labelled "Complete Markets, prod. shock only" report business cycle statistics for the model with complete financial markets and productivity shocks only; the rows labelled "Complete markets, both shocks" refers to the model with complete markets and both productivity shocks and shocks to the efficiency of technology capital; the rows labelled "Incomplete markets, both shocks" refer to the incomplete markets case with both shocks.

There are three notable results. First, all model versions generates business cycle statistics similar to those of standard international business cycle models which includes the well known short-comings. The complete market models imply perfect cross-country correlation of consumption, negative cross-country employment correlations, and pro-cyclical net exports. Interestingly, the complete market models generate volatilities of net exports in line with the data, while in the incomplete market model volatility of net exports is too low compared to the data.

Second, and quiet interestingly, the cross-country correlation of consumption under incomplete markets is equal to the correlation of GDP. This means the model can to some extend address the so called "quantity anomaly", that is, the fact that in the standard business cycle model consumption correlations are more positive than GDP correlations while in the data the opposite is true. One reason for this finding is the measurement issue for GDP. Recall that GDP is value added minus investment in technology capital. We will see below that when countries are relatively closed, measured GDP over-estimates the cross-country correlation of aggregate activity. This means that cross-country correlation of actual output is lower. In addition, when countries are relatively closed to FDI, cross-country correlations of consumption is closer to actual value added. The other reason is the presence of multinational-specific shocks that are uncorrelated across countries. Absent other international assets than FDI, households cannot insure this risk and consumption correlations are lower than without these shocks. I conjecture that for the same reasons the incomplete markets model implies a cross-country correlation of employment in line with the data.

Third, the introduction of multinational-specific shocks leaves the business cycle moments mostly unaffected while it helps to match the volatility of measured returns. This confirms to some extent the validity of my calibration strategy where the volatility of the multinational-specific shocks was chosen in order to match the standard deviation of measured returns on FDI.

To summarize, the results in this section show that introducing technology capital and multinational-specific shocks into a standard international business cycle model generates plausible business cycle statistics, and helps to explain some features of the data the standard model has difficulties with.

4.2 FDI integration and business cycle synchronization

This sections connects the quantitative results of the model with the empirical results in the first part of the paper. I start by varying the FDI openness parameter τ and discuss how FDI openness affects co-movement of investment and GDP, respectively. Second, using artificial data generated by the model, I run the same regressions as in the data and compare the obtained regression coefficients. This is a simple test whether the causal relation in the model is consistent with the data. In addition, the model allows me to distinguish between measured GDP as reported in national accounts and actual value added and I will show how their co-movement patterns differ.

4.2.1 FDI openness and investment synchronization

For each model version as described above (complete markets with productivity shocks only, complete markets with both shocks, and incomplete markets with both shocks), I



Figure 3: FDI openness and investment synchronization

vary the degree of FDI openness from very low integration ($\tau = 0.05$) to complete FDI openness ($\tau = 1$).³³ For each value of the financial integration parameter I report the average synchronization of investment as defined in equation (2). Figure 3 shows the results of this exercise. In all model versions, the slope of the line is always positive; a higher degree of FDI openness leads to more correlated investment cycles. This result is consistent with my regression estimates in table 2.

There are three main conclusions from the comparison of the three model versions. First, there is a clear ranking in terms of the slope of the increase in investment synchronization: the smallest in the model with complete markets and productivity shocks only (panel a))

³³ I take a value for τ that is slightly bigger than zero for numerical reasons; a value of $\tau = 0.05$ implies a bilateral FDI to GDP ratio of 0.015 percent, i.e. the 2.5 percent quantile in the data.

and the strongest in complete markets and both productivity and technology capital shocks (panel b)). This suggest that both technology capital and multinational-specific shocks are quantitatively important for understanding investment co-movement.

Second, the slope of the increase in investment synchronization under incomplete markets (panel c)) lies in between the two models with complete markets.³⁴ When markets are incomplete (recall that FDI is assumed to be the only asset) and both shocks are active, there are two counter-acting forces: on the one hand, in response to countryspecific productivity shocks, investment tends to flow to the more productive country and this weakens investment co-movement; on the other hand, when openness increases, firm-specific shocks to the efficiency of technology capital become more important and returns on investment within firm become more correlated. As described in section 3.5.2 this strengthens investment co-movement. As is evident in panel c), the latter force is dominating for most of the range of FDI openness τ .

Third, in all model versions, the increase in investment synchronization is fastest for middle ranges of FDI openness τ between 0.2 and 0.8; the reason behind this is the fact that, for these values of τ , the gains from FDI are biggest and FDI positions increase the fastest, as shown in panel d).

4.2.2 FDI openness and GDP synchronization.

Let us now turn to the synchronization patterns of GDP as measured in national accounts (equation (20)) and actual value added (equation (7)). Again, for each model version (complete markets with productivity shocks only, complete markets with both shocks, and incomplete markets with both shocks), I vary the degree of FDI openness from very low integration ($\tau = 0.05$) to complete FDI openness ($\tau = 1$).³⁵ For each value of the financial integration parameter I report the average synchronization of GDP and value added as defined in equation (1). The results are shown in figure 4.

In all model versions, synchronization patterns of measured GDP (blue solid lines) and

³⁴When considering incomplete markets with productivity shocks only, investment synchronization responds slightly negative in response to variation in τ (not shown), see appendix.

 $^{^{35}}$ See footnote 33.



Figure 4: FDI openness and GDP synchronization

value added (dashed red lines) behave quiet differently: while GDP shows a non-linear pattern in response to variations to FDI openness τ , synchronization of value added is constant (panel a)) or monotone increasing (panels b) and c)). This means that - due to the mismeasurement in GDP - actual cross country co-movement is overestimated for relatively closed country-pairs and underestimated when countries are relatively open to FDI.

We can gain some intuition by answering the following two questions. First, under complete markets with productivity shocks only (panel a)), why is GDP synchronization falling when FDI opennes increases? When FDI openness increases, a positive productivity shock in one country benefits multinationals from both countries due to their increased operations abroad. On impact, because expected returns increase, multinationals from both countries increase their investment in technology capital. The more the countries are open, the more similar in terms of size is this increase in investment in technology capital. Domestic investment in technology capital increases by relatively less and foreign investment in technology capital increases by relatively more when comparing a relatively open country pair to a relatively closed country pair. Everything else equal, by the definition of GDP (value added minus investment in technology capital), measured GDP at home is then relatively higher and GDP abroad is relatively lower; GDP is more negatively correlated.

Second, when looking at panels b) and c), why is GDP synchronization first increasing (for values of τ below 0.6) and then falling again (for values of τ above 0.6)? Consider first the complete market case as shown in panel b). In this model version, both country-specific and multinational-specific shocks are active. In the range of τ between 0.2 and 0.6 the returns of increasing FDI is biggest as discussed above and also shown in panel d). Therefore in this range, the country-specific and multinational-specific shock work in the same way and measured GDP resembles the synchronization pattern of value added (while the correlation of investment in technology capital is also increasing, see figure 1). Recall that following a positive shock to the domestic multinational, value added of both countries increases because the multinational invests in both countries more. At τ equal to approximately 0.6, however, there is a dipping point. Multinational-specific shocks dominate and investments in technology capital become negatively correlated, even exceeding the increase in correlation of value added. By the definition of GDP (value added minus investment in technology capital), GDP correlation therefore decreases.

Finally, under incomplete markets, the pattern of GDP and value added synchronization resembles the pattern under complete markets. However, the lines in panel c are flatter than the ones in panel b). Incomplete markets dampen the amplitude of the synchronization of GDP and value added by the same amount. This is because under incomplete markets, this comparative static exercise is not really 'ceteris paribus'. Besides the effects of technology flows associated with more FDI as described under complete markets, varying the parameter τ picks up another force: the possibility of countries to shift resources across country borders, hence completing the markets in terms of risk sharing. In this

	Data	Model		
		Complete Markets, prod. only	Complete markets, both shocks	Incomplete markets, both shocks
Coefficient on investment synchronization	4.106 (2.48)	0.186	0.831	0.258
Coefficient on GDP synchronization	0.133 (0.46)	-0.092	0.159	0.022
Coefficient on synchron. of value added	n.a. n.a.	-0.001	0.261	0.062

Table 7: Bilateral FDI linkages and synchronization: data vs. model

Notes: The table reports the estimated coefficients in the three model versions. For convenience, column one reports the estimated coefficients obtained from the data, taken from column (3) in tables 2 and 3, respectively. For the empirical estimates, t-values are reported in parenthesis below. The last three columns report the estimated coefficient of the three model versions. In the first row, the dependent variable is investment synchronization, in the second row GDP synchronization, and in the third row it is actual value added (Y). The right wave variable in all regressions is log of the sum of bilateral FDI positions divided by the sum of the countries' GDP.

sense for each value of τ the agents face a different international market structure and the effect of FDI on investment and output co-movement is downward biased.

4.2.3 Regression coefficients in the model and data.

In this section, I run similar regressions on model data to the ones in the empirical part of the paper. In particular, I simulate the model for ten country pairs, varying the FDI openness parameter (τ) smoothly from zero to one. For each country pair I simulate the model for 48 quarters (12 years as in the data) and construct the same measure of Investment and GDP synchronization and for bilateral FDI linkages as used in the data analysis. I then convert the quarterly data to the yearly frequency by averaging across quarters. Finally, I regress the obtained synchronization measures on the log of bilateral FDI linkages. Table 7 reports the results for the three model versions considered. Note that, in the last row of the table, I report the synchronization measure for actual value added; in the model value added differs from measured GDP because investment in technology capital is expensed in national accounts. In the data, I do not have an equivalent measure available, for this reason I put "n.a." in the respective column. For comparison in table 7 I also report the coefficients on the same regression using actual data, repeating the estimates in column (3) in tables 2 and 3, respectively.

Three results emerge. First, I find that overall a higher degree of FDI integration leads to higher level of investment synchronization, as reflected in an integration coefficient between 0.186 and 0.831. Therefore the model explains up to 20 percent of the estimated coefficient in the data (4.106).

Second, the the sign of relation between FDI integration and GDP synchronization is ambiguous; the estimated coefficients in the model range from -0.092 to 0.159 where the latter value is very close to the coefficient obtained from actual data (0.129). This suggests that the relation between financial integration and output co-movement implied by our model is statistically close to the one we estimate in the data. In terms of magnitude, the estimated coefficients for GDP synchronization are significantly smaller than the coefficients obtained for investment synchronization, as observed in the data.

Third, as reported in the last row of table 7, actual co-movement of aggregate activity is larger than suggested by the estimates for GDP synchronization: the regression coefficients on actual value added range from -0.001 to 0.261, always lying above the estimates for GDP synchronization. This means that focussing on measured GDP as an indicator for co-movement underestimates the actual business cycle co-movement. To get a sense for the magnitudes, consider the estimated coefficients for the model version with complete markets and both shocks, reported in column two. The coefficient for GDP synchronization is 0.159 while the one for actual value added is 0.261. Hence, when focussing on GDP, a doubling of FDI linkages would lead to an increase in GDP synchronization of 0.159 percentage points. In contrast, for value added a doubling of FDI linkages leads to an increase in synchronization of 0.261 percentage points, a value that is 65 percent higher than the one for GDP synchronization. Therefore the model gives a theoretical rationale for a substantial bias due to measurement error that emerges because national accounts do not include intangible capital.
5 Summary

In this paper, I document that FDI integration and investment synchronization are positively correlated. This is an important finding because the standard international business cycle model predicts the opposite. I then propose a tractable international business cycle model where multinational firms engage in FDI and multinational activity is subject to shocks. I show that the positive association between FDI openness and investment synchronization is consistent with the hypothesis that multinationals play an important role for the international transmission of shocks. The model also gives a rationale for the weak link between financial integration on GDP co-movement. Due to mismeasurement of GDP reported in national accounts, actual output co-movement is overestimated when countries are relatively closed and underestimated when countries are relatively open to FDI.

There are three main lessons from the theory. First, more FDI openness leads unambiguously to higher investment synchronization across country borders. This result holds in model specifications with complete or incomplete financial markets, respectively.

Second, shocks to multinational activity are important to quantitatively account for the estimated link between FDI integration and investment synchronization. The regression coefficients estimated on artificial data from the model with multinational-specific shocks quantitatively explain 20 percent of the empirical regression coefficients, compared to 6 percent in the model version without multinational-specific shocks.

Third, because measured GDP is distorted, the regression coefficients on actual output synchronization are 0.04 to 0.1 percentage points higher than the coefficients for GDP synchronization. This are significant numbers amounting to 30 to 75 percent of the empirical point estimate. As I abstract from many other forms of intangible capital (potentially worsen the measurement issue) it is not clear whether these estimates are upper or lower bounds. I leave this for future research.

References

- ALVAREZ, F., AND R. E. LUCAS (2007): "General equilibrium analysis of the Eaton– Kortum model of international trade," *Journal of monetary Economics*, 54(6), 1726– 1768.
- ATKESON, A., AND P. J. KEHOE (2005): "Modeling and Measuring Organization Capital," *Journal of Political Economy*, 113(5), pp. 1026–1053.
- BACKUS, D. K., P. J. KEHOE, AND F. E. KYDLAND (1992): "International real business cycles," *Journal of Political Economy*, pp. 745–775.
- BAXTER, M., AND M. J. CRUCINI (1995): "Business Cycles and the Asset Structure of Foreign Trade," *International Economic Review*, pp. 821–854.
- BAXTER, M., AND M. A. KOUPARITSAS (2005): "Determinants of business cycle comovement: a robust analysis," *Journal of Monetary Economics*, 52(1), 113–157.
- EISFELDT, A. L., AND D. PAPANIKOLAOU (2013): "Organization Capital and the Cross-Section of Expected Returns," *The Journal of Finance*.
- FRANKEL, J. A., AND A. K. ROSE (1998): "The endogenity of the optimum currency area criteria," *The Economic Journal*, 108(449), 1009–1025.
- HEATHCOTE, J., AND F. PERRI (2002): "Financial autarky and international business cycles," *Journal of Monetary Economics*, 49(3), 601–628.
- HEATHCOTE, J., AND F. PERRI (2004): "Financial globalization and real regionalization," Journal of Economic Theory, 119(1), 207–243.
- HSU, C.-C., J.-Y. WU, AND R. YAU (2011): "Foreign direct investment and business cycle co-movements: The panel data evidence," *Journal of Macroeconomics*, 33(4), 770 783.
- IMBS, J. (2004): "Trade, finance, specialization, and synchronization," Review of Economics and Statistics, 86(3), 723-734.

(2006): "The real effects of financial integration," Journal of International Economics, 68(2), 296–324.

- JOHRI, A., M.-A. LETENDRE, AND D. LUO (2011): "Organizational capital and the international co-movement of investment," *Journal of Macroeconomics*, 33(4), 511–523.
- KALEMLI-OZCAN, S., E. PAPAIOANNOU, AND F. PERRI (2013): "Global banks and crisis transmission," Journal of International Economics, 89(2), 495 510.
- KALEMLI-OZCAN, S., E. PAPAIOANNOU, AND J.-L. PEYDRÓ (2013): "Financial Regulation, Financial Globalization, and the Synchronization of Economic Activity," The Journal of Finance, 68(3), 1179–1228.
- KAPICKA, M. (2012): "How Important is Technology Capital for the United States?," American Economic Journal: Macroeconomics, 4(2), 218–248.
- KEHOE, P. J., AND F. PERRI (2002): "International business cycles with endogenous incomplete markets," *Econometrica*, 70(3), 907–928.
- KOSE, M., AND K. YI (2006): "Can the standard international business cycle model explain the relation between trade and comovement?," *Journal of International Eco*nomics, 68(2), 267–295.
- LANE, P., AND G. MILESI-FERRETTI (2007): "The external wealth of nations mark II: Revised and extended estimates of foreign assets and liabilities, 1970-2004," *Journal of International Economics*, 73(2), 223-250.
- LUSTIG, H., C. SYVERSON, AND S. VAN NIEUWERBURGH (2011): "Technological change and the growing inequality in managerial compensation," *Journal of Financial Economics*, 99(3), 601–627.
- MCGRATTAN, E., AND E. PRESCOTT (2009): "Openness, technology capital, and development," *Journal of Economic Theory*, 144(6), 2454–2476.

^{— (2010): &}quot;Technology Capital and the US Current Account," *The American Economic Review*, 100(4), 1493–1522.

- OUYANG, M. (2011): "On the Cyclicality of R&D," *Review of Economics and Statistics*, 93(2), 542–553.
- PERRI, F., AND V. QUADRINI (2011): "International Recessions," Working Paper 17201, National Bureau of Economic Research.
- QUADRINI, V., AND U. J. JERMANN (2012): "Macroeconomic Effects of Financial Shocks," American Economic Review, 102(1), 238–71.
- RAMONDO, N., AND V. RAPPOPORT (2010): "The role of multinational production in a risky environment," *Journal of International Economics*, 81(2), 240 252.
- WÄLDE, K., AND U. WOITEK (2004): "R&D expenditure in G7 countries and the implications for endogenous fluctuations and growth," *Economics Letters*, 82(1), 91–97.

A Data

Quarterly output and investment data are from OECD Quarterly National Accounts. The real series are annualized in constant US Dollars (OECD reference year), converted at fixed PPP exchange rates and seasonally adjusted (series VPVOBARSA). Correlations are the yearly average of 40 quarter rolling window estimates after HP-filtering data. The synchronization measures are calculated as explained in the main text. Yearly nominal GDP is from OECD National Accounts, annualized values in current US Dollars, converted at current PPP exchange rates, and seasonally adjusted (series CPCARSA). Bilateral outward and inward foreign direct investment data are from the OECD Foreign Direct Investment Database at a yearly frequency. Note that a general caveat of FDI data is that it is usually recorded at historical cost, so the FDI positions do not necessarily reflect actual market values. Total foreign direct investment data are from the OECD foreign direct investment data and from Lane and Milesi-Ferretti (2007). Bilateral exports and imports are from the OECD monthly trade statistics. Population data are from OECD labor market statistics.

Table 8 summarizes the country pairs with all FDI data are available throughout the

whole sample from 1990 to 2006 and 1995 to 2006, respectively. An exception is Japan, bilateral FDI data are available from 1985 to 1994 and 1996 to 2006 only, 1995 is not reported in the OECD database. For not deleting a G7 member from the analysis, I used linear interpolation to obtain the bilateral FDI positions for 1995.

Balanced Sample 19	95 - 2006 used i	in tables 2 and	13				
AUT,CAN	AUT,DEU	AUT,FRA	AUT,GBR	AUT,NLD	AUT,USA	CAN,DEU	CAN,FRA
CAN,GBR	CAN, ITA	CAN, JPN	CAN, NLD	CAN,NOR	CAN, USA	DEU,GBR	DEU,ITA
DEU, JPN	DEU,NLD	DEU,NOR	DEU,SWE	DEU,USA	FIN,DEU	FIN,FRA	FIN,GBR
FIN,NLD	FIN,NOR	FIN,SWE	FIN, USA	FRA,DEU	FRA,GBR	FRA,ITA	FRA, JPN
FRA,NLD	FRA,NOR	FRA,SWE	FRA, USA	$_{\rm GBR, USA}$	$_{\rm ITA,GBR}$	ITA, JPN	ITA,NLD
ITA,USA	$_{\rm JPN,GBR}$	$_{\rm JPN,NLD}$	JPN,USA	$_{\rm NLD,GBR}$	NLD,NOR	NLD,SWE	NLD,USA
NOR,SWE	$_{ m NOR,USA}$	PRT,GBR	$_{\rm SWE,GBR}$	$_{\rm SWE, USA}$			
Balanced Sample fo	r hp-filtered dat	a using 5-year	time intervals	, 1991 - 1996,	1996 - 2001, 2	001-2006, in	tables 11 and 12
AUT,CAN	AUT,DEU	AUT,FRA	AUT,GBR	AUT,USA	CAN,DEU	CAN,FRA	CAN,GBR
CAN,ITA	CAN, JPN	CAN,NOR	CAN, USA	DEU,GBR	DEU,ITA	DEU,JPN	DEU,NLD
DEU,NOR	DEU,SWE	DEU,USA	FIN,DEU	FIN,FRA	FIN,GBR	FIN,NOR	FIN, USA
FRA, DEU							
	FRA,GBR	FRA,ITA	FRA, JPN	FRA,NLD	FRA,NOR	FRA,SWE	FRA, USA
GBR,USA	$_{ m FRA,GBR}$ ITA,GBR	FRA,ITA ITA,JPN	FRA,JPN ITA,NLD	FRA,NLD ITA,USA	FRA,NOR JPN,GBR	FRA,SWE JPN,NLD	FRA,USA JPN,USA
${ m GBR, USA} { m NLD, GBR}$	· · · · · · · · · · · · · · · · · · ·	/	· · · · ·	· · · · · ·	,	· · · · · ·	/

Table 8: Country pairs used in empirical analysis

Notes: The table lists the country-pairs for which bilateral data were available in all years without gaps used in the respective samples. Essentially, it consists of the 21 G7 country pairs plus pairs involving Austria, Finland, Netherlands, Norway, Portugal, and Sweden. Please read the note regarding the bilateral FDI positions reported by Japan in this appendix.

B Additional empirical results and robustness checks

This section conducts several robustness checks for the empirical results in the main text. First, I restrict the analysis for G7 countries only between 1991 to 2006. Table 9 reports the results for investment synchronization. The main message remains, more bilateral FDI linkages are associated with more investment synchronization. Note, however, that the point estimates are somewhat lower, equal to 2.57 in column (3) of table 9 versus 4.1 in column (3) of table 2 in the main text. The results for GDP synchronization are

Second, I split the sample in three non-overlapping time periods, 1991-1996, 1996-2001, and 2001 - 2006. As dependent variables I use the cross-country correlation of hp-filtered GDP and investment, respectively, computed for the 20 quarters of each time period. I then regress GDP and investment correlation on the log of bilateral FDI linkages at the beginning of each time period, that is, FDI position in 1991, 1996, and 2001, respectively. Table 11 and table 12 report the results of this exercise. The findings are similar to the ones of the benchmark estimates in the main text. More FDI linkages are associated with higher investment correlations, and there is no statistical link between FDI linkages and GDP correlations.

 Table 9: Bilateral FDI linkages and investment synchronization, G7 country-pairs only

 Dependent Variable: Investment growth synchronization (annualized)

	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
FDI/GDP	3.321^{**} (2.83)	1.984^{*} (1.84)	2.135^{*} (1.86)	2.175^{*} (1.88)				
$\mathrm{Trade}/\mathrm{GDP}$	× ,	· · /	· · /	-1.088 (-0.26)				
FDI/Total FDI				()	1.754^{*}	1.879^{*}	1.960*	2.038^{*}
Trade/Total Trade					(1.91)	(1.88)	(1.91)	(1.91) -1.549 (-0.40)
Country-pair fixed	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Time fixed	No	Yes	Yes	Yes	No	Yes	Yes	Yes
Country trends	Yes	No	Yes	Yes	Yes	No	Yes	Yes
R-squared (within)	0.0976	0.234	0.277	0.277	0.0841	0.234	0.276	0.277
Observations	294	315	294	294	294	315	294	294

Notes: The table reports panel (country-pair) fixed-effect coefficients estimated over the period 1991 to 2006 of the G7 countries (21 country pairs). The dependent variable is minus one times the absolute value of the difference in quarterly growth rate of aggregate investment between country *i* and *j* in year *t* (the yearly estimate is obtained by averaging over the respective four quarterly estimates). In columns (1) - (4) FDI integration is measured by the log of the share of the stock of bilateral Foreign Direct Investment Assets and Liabilities between countries *i* and *j* in the previous year relative to the sum of the two countries' GDP in the previous year (denoted FDI/GDP). In columns (5) - (8) FDI integration is measured by the log of the share of the stock of bilateral Foreign Direct Investment Assets and Liabilities between countries *i* and *j* in the previous year relative to the sum of the two countries' total FDI Assets and FDI Liabilities in the entire world in the previous year relative to the sum of the two countries' does the countries' GDP and the log of the product of the two countries *i* and *j* in the previous year. The specification in (4) includes the log of the share of bilateral export and import flows between countries *i* and *j* in the previous year relative to the sum of the two countries (3 GDP in the previous year (Trade/GDP). The specification (8) includes the log of the share of bilateral export and import flows between countries *i* and *j* in the previous and imports in the previous year (Trade/Total Trade). The specifications in columns (3), (4), (7), and (8) include time fixed-effects. The specifications in columns (3), (4), (7), and (8) include time fixed-effects and country-specific linear time-trends. Standard errors adjusted for panel (country-pair) specific auto-correlation and heteroskedasticity and corresponding t-statistics are reported below. * denotes significance at the 90% confidence level, *** denotes significance at the 99% confidence level. For a det

C Model details and additional results for the incomplete market economy

C.1 Model Equations

In this subsection, I list the model equations, including the first order conditions of households and firms. Note that because the model is symmetric, the first order conditions for foreign firms are omitted.

	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
FDI/GDP	$0.430 \\ (1.46)$	-0.0668 (-0.20)	-0.0791 (-0.23)	-0.0538 (-0.15)				
$\mathrm{Trade}/\mathrm{GDP}$	(1110)	(0.20)	(0.20)	-0.683 (-0.91)				
FDI/Total FDI				· · ·	$\begin{array}{c} 0.314 \\ (1.04) \end{array}$		-0.0874 (-0.25)	-0.0956 (-0.27)
Trade/Total Trade								$0.163 \\ (0.26)$
Country-pair fixed	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Time fixed	No	Yes	Yes	Yes	No	Yes	Yes	Yes
Country trends	Yes	No	Yes	Yes	Yes	No	Yes	Yes
R-squared (within)	0.0413	0.230	0.281	0.282	0.0393	0.187	0.281	0.281
Observations	294	315	294	294	294	315	294	294

Table 10: Bilateral FDI linkages and GDP synchronization, G7 country-pairs only

Notes: The table reports panel (country-pair) fixed-effect coefficients estimated over the period 1991 to 2006 of the G7 countries (21 country pairs). The dependent variable is minus one times the absolute value of the difference in quarterly growth rate of aggregate investment between country *i* and *j* in year *t* (the yearly estimate is obtained by averaging over the respective four quarterly estimates). In columns (1) - (4) FDI integration is measured by the log of the share of the stock of bilateral Foreign Direct Investment Assets and Liabilities between countries *i* and *j* in the previous year relative to the sum of the two countries' GDP in the previous year (denoted FDI/GDP). In columns (5) - (8) FDI integration is measured by the log of the share of the stock of bilateral Foreign Direct Investment Assets and Liabilities between countries *i* and *j* in the previous year relative to the sum of the two countries' total FDI Assets and FDI Liabilities in the entire world in the previous year (denoted FDI/Total FDI). All specifications also include the log of the share of bilateral export and import flows between countries *i* and *j* in the previous year relative to the sum of the two countries' GDP in the previous year (Trade/GDP). The specification in (4) includes the log of the share of bilateral export and import flows between countries *i* and *j* in the previous year relative to the sum of the two countries' GDP in the previous year (Trade/GDP). The specifications (8) includes the log of the share of bilateral export and import flows between countries *i* and *j* in the previous year metative to the sum of the two countries *i* and *j* in the previous year relative to the sum of the two countries *i* and *j* in the previous year relative to the sum of the two countries *i* and *j* in the previous year relative to the sum of the two countries *i* and *j* in the previous year relative to the sum of the two countries *i* and *j* in the previous and import flows between countries *i* and *j* in the Notes: The table reports panel (country-pair) fixed-effect coefficients estimated over the period 1991 to 2006 of the G7 countries (21 country appendix A.

Table 11: Bilateral	FDI linkages and	investment co	orrelations. h [.]	p-filtered data

	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
FDI/GDP	0.548^{**} (2.65)	0.335^{**} (2.46)	0.578^{**} (2.09)	0.654^{**} (2.11)				
$\mathrm{Trade}/\mathrm{GDP}$	· · · ·	· · ·	· · ·	-0.919 (-1.06)				
FDI/Total FDI				· · ·	0.585^{*} (1.86)	0.325^{**} (2.17)	0.546^{*} (1.71)	0.680^{*} (1.88)
Trade/Total Trade					(1100)	(2.11)	(1.1.1)	(-1.739) (-1.51)
Country-pair fixed	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Time fixed	No	Yes	Yes	Yes	No	Yes	Yes	Yes
Country trends	Yes	No	Yes	Yes	Yes	No	Yes	Yes
R-squared (within)	0.126	0.286	0.127	0.150	0.105	0.279	0.112	0.170
Country-pairs	46	46	46	46	46	46	46	46
Observations	92	138	92	92	92	138	92	92

Dependent Variable: cross-country correlation of hp-filtered investment

Notes: The table reports panel (country-pair) fixed-effect coefficients estimated over three non-overlapping time intervals, 1991 - 1996, 1996 - 2001, and 2001 - 2006, using a balanced panel for 42 country-pairs for which bilateral FDI positions are available for all years. A list of included countries-pairs can be found in table ??. The dependent variable is the five-year average cross-country correlation of hp-filtered investment in country i and j for each time interval. In columns (1) - (4) FDI integration is measured by the log of the share of the stock of bilateral Foreign Direct Investment Assets and Liabilities between countries i and j in the initial year of the respective time period relative to the sum of the two countries' GDP in the in the initial year of the respective time period (denoted FDI/GDP). In columns (5) - (8) FDI integration is measured by the log of the share of the stock of bilateral Foreign Direct Investment Assets and FDI Liabilities vis-a-vis the entire world in the initial year of the respective time period (denoted FDI/Total FDI). All specifications also include the log of the two countries' GDP and the log of the product of the two countries' population in the initial year of the respective time period. Specification (4) includes the log of the share of bilateral export and import flows between countries i and j in the initial year of the respective time period relative to the sum of the two countries' GDP in the initial year of the respective time period relative to the sum of the two countries' GDP in the initial year of the respective time period (Trade/GDP). Specification (8) includes the log of the share of bilateral export and imports in the initial year of the respective time period relative to the sum of the two countries's and imports in the initial year of the respective time period (Trade/GDP). Specifications in columns (1) and (5) include country-specific linear time-trends. The specifications in columns (2) and (6) include time fixed-effects. The specifications in The table reports panel (country-pair) fixed-effect coefficients estimated over three non-overlapping time intervals, 1991 - 1996, 1996 detailed data description see appendix A

Dependent Variable: cross-country correlation of hp-filtered GDP									
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	
FDI/GDP	-0.309 (-1.02)	$0.184 \\ (1.05)$	-0.0581 (-0.18)	-0.0862 (-0.28)					
$\operatorname{Trade}/\operatorname{GDP}$				$0.343 \\ (0.54)$					
FDI/Total FDI				· · ·	-0.0509	$0.213 \\ (1.19)$	$0.0679 \\ (0.21)$	$0.161 \\ (0.48)$	
Trade/Total Trade					()	· · ·	. ,	(-1.215)	
Country-pair fixed	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	
Time fixed	No	Yes	Yes	Yes	No	Yes	Yes	Yes	
Country trends	Yes	No	Yes	Yes	Yes	No	Yes	Yes	
R-squared (within)	0.382	0.236	0.417	0.420	0.362	0.238	0.417	0.440	
Country-pairs	46	46	46	46	46	46	46	46	
Observations	92	138	92	92	92	138	92	92	

Table 12: Bilateral FDI linkages and output correlations, hp-filtered data

Observations921389292921389292Notes: The table reports panel (country-pair) fixed-effect coefficients estimated over three non-overlapping time intervals, 1991 - 1996, 19962001, and 2001 - 2006, using a balanced panel for 42 country-pairs for which bilateral FDI positions are available for all years. A list of
investment in country i and j for each time interval. In columns (1) - (4) FDI integration is measured by the log of the share of the stock of
to the sum of the two countries' GDP in the in the initial year of the respective time period (denoted FDI/GDP). In columns (5) - (8) FDI
integration is measured by the log of the share of the stock of bilateral Foreign Direct Investment Assets and Clabilities visa-eits
the entire world in the initial year of the respective time period (denoted FDI/GDP). In columns (5) - (8) FDI
includes the log of the product of the two countries' population in the initial year of the respective time period.
Specifications also include the log of the share of bilateral export and import flows between countries i and j in the initial year of the respective time period
relative to the sum of the two countries' GDP in the initial year of the respective time period (Trade/GDP). Specification (8) includes the
log of the share of bilateral export and import flows between countries i and j in the initial year of the respective time period
relative to the sum of the two countries' GDP in the initial year of the respective time period (Trade/GDP). Specification (8) includes the
specifications in columns (3), (4), (7), and (8) include time fixed-effects and country-specific linear time-trends. Standard errors adjusted for
specifications in columns (3), (4), (7), and (8) include time fixed-effects and country-specific linear time-trends. Standard errors adjusted for
at the 90% confidence level, ** denotes significance at

Output in tradable good sector in the home country and relative price of intermediate good bundle

$$Y_t = A_t (k_{Tt}^{\theta} l_{Tt}^{1-\theta})^{\nu} X^{1-\nu}$$
$$P_t = (1-\nu) \frac{Y_t}{X_t}$$

Output on intermediate good sector in the home country

$$Y_{It} = \left[y_{dt}^{\eta} + (y_{ft}^*)^{\eta} \right]^{\frac{1}{\eta}}$$

Optimality conditions w.r.t. labor for home firms

$$0 = (1 - \theta)\nu Y_t - w_t l_{Tt}$$

$$0 = (1 - \theta)(1 - \phi)\eta P_t Y_{It}^{1 - \eta} y_{dt}^{\eta} - w_t l_{dt}$$

$$0 = (1 - \theta)(1 - \phi)\eta P_t^* (Y_{It}^*)^{1 - \eta} y_{ft}^{\eta} - w^* t l_{ft}$$

Value of an additional unit of investment (i.e. Lagrange multiplier on investment accumulation equation) for home firms

$$V_{Tt} = \left[\chi_1 \left(\frac{i_{Tt}}{k_{Tt-1}}\right)^{-\psi}\right]^{-1}$$
$$V_{dt} = \left[\chi_1 \left(\frac{i_{dt}}{k_{dt-1}}\right)^{-\psi}\right]^{-1}$$
$$V_{ft} = \left[\chi_1 \left(\frac{i_{ft}}{k_{ft-1}}\right)^{-\psi}\right]^{-1}$$
$$V_{Mt} = \left[\chi_1^m \left(\frac{i_{Mt}}{M_{t-1}}\right)^{-\psi_m}\right]^{-1}$$

Optimality conditions with respect to investment, home firms

$$\begin{aligned} V_{Tt}k_{Tt+1} &= \beta E_t \left\{ \frac{\lambda_{t+1}}{\lambda_t} \left(\theta \eta Y_{t+1} - i_{Tt+1} + V_{Tt+1} k_{Tt+2} \right) \right\} \\ V_{dt}k_{dt+1} &= \beta E_t \left\{ \frac{\lambda_{t+1}}{\lambda_t} \left(\theta (1-\phi) \eta P_t Y_{It}^{1-\eta} y_{dt}^{\eta} - i_{dt+1} + V_{dt+1} k_{dt+2} \right) \right\} \\ V_{ft}k_{ft+1} &= \beta E_t \left\{ \frac{\lambda_{t+1}}{\lambda_t} \left(\theta (1-\phi) \eta P_t^* (Y_{It}^*)^{1-\eta} y_{ft}^{\eta} - i_{ft+1} + V_{ft+1} k_{ft+2} \right) \right\} \\ V_{Mt}M_{t+1} &= \beta E_t \left\{ \frac{\lambda_{t+1}}{\lambda_t} \left[\phi \eta \left(P_t Y_{It}^{1-\eta} y_{dt}^{\eta} + P_t^* (Y_{It}^*)^{1-\eta} y_{ft}^{\eta} \right) - i_{Mt+1} + V_{Mt+1} M_{t+2} \right] \right\} \end{aligned}$$

Domestic and foreign production by home owned multinationals

$$y_{dt} = e^{z_t} M_t^{\phi} \left(k_{dt}^{\theta} l_{dt}^{1-\theta} \right)^{1-\phi}$$
$$y_{ft} = \tau e^{z_t} M_t^{\phi} \left(k_{ft}^{\theta} l_{ft}^{1-\theta} \right)^{1-\phi}$$

Total dividends households in the home country receive

$$d_t = (Y_t - w_t l_{Tt} - i_{Tt}) + (P_t Y_{It}^{1-\eta} y_{dt}^{\eta} - w_t l_{dt} - i_{dt}) + (P_t^* (Y_{It}^*)^{1-\eta} y_{ft}^{\eta} - w_t^* l_{ft} - i_{ft}) - i_{Mt}$$

Capital accumulation

$$k_{jt+1} = (1-\delta)k_{jt} + \left[\frac{\chi_1\left(\frac{i_{jt}}{k_{jt}}\right)^{1-\psi}}{1-\psi} + \chi_2\right]k_{jt} \qquad j = T, d, f$$
$$M_{t+1} = (1-\delta_m)M_t + \left[\frac{\chi_1^m\left(\frac{i_{Mt}}{M_t}\right)^{1-\psi_m}}{1-\psi_m} + \chi_2^m\right]M_t$$

Analoguous for foreign firms. Domestic and foreign households

$$\lambda_{t} = U_{1}(C_{t}, \bar{L} - L_{t})$$

$$\lambda_{t}^{*} = U_{1}(C_{t}^{*}, \bar{L} - L_{t}^{*})$$

$$0 = U_{2}(C_{t}, \bar{L} - L_{t}) + \lambda_{t}w_{t}$$

$$0 = U_{2}(C_{t}^{*}, \bar{L} - L_{t}^{*}) + \lambda_{t}^{*}w_{t}^{*}$$

Labor market clearing

$$L_t = l_{Tt} + l_{dt} + l_{ft}^*$$
$$L_t^* = l_{Tt}^* + l_{dt}^* + l_{ft}$$

Goods market clearing

$$Y_t + Y_t^* = C_t + C_t^* + i_{Tt} + i_{Tt}^* + i_{dt} + i_{ft} + i_{dt}^* + i_{ft}^* + i_{Mt} + i_{Mt}^*$$

International financial markets:

a) Complete financial markets

$$\lambda_t = \lambda_t^*$$

b) Incomplete financial markets

$$C_t = w_t L_t + d_{Tt} + d_{Mt}$$

C.2 Deterministic steady state

The system of equations in the deterministic steady state for two complete open economies $\tau = 1$ are given by

$$\frac{1 - \beta(1 - \delta)}{\nu\theta\beta} = \frac{Y}{K_1} \qquad \qquad \frac{1 - \beta(1 - \delta)}{\nu\theta\beta} = \frac{Y^*}{K_1^*} \\
\frac{1 - \beta(1 - \delta)}{(1 - \phi)\theta\eta\beta} = PY_2^{1 - \eta}\frac{y_d^{\eta}}{k_d} \qquad \qquad \frac{1 - \beta(1 - \delta)}{(1 - \phi)\theta\eta\beta} = PY_2^{1 - \eta}\frac{(y_f^*)^{\eta}}{k_f^*} \\
\frac{1 - \beta(1 - \delta)}{(1 - \phi)\theta\eta\beta} = P^*(Y_2^*)^{1 - \eta}\frac{y_f^{\eta}}{k_f} \qquad \qquad \frac{1 - \beta(1 - \delta)}{(1 - \phi)\theta\eta\beta} = P^*(Y_2^*)^{1 - \eta}\frac{(y_d^*)^{\eta}}{k_d^*} \\
\frac{1 - \beta(1 - \delta_m)}{\phi\eta\beta} = \frac{PY_2^{1 - \eta}y_d^{\eta} + P^*(Y_2^*)^{1 - \eta}y_f^{\eta}}{M} \qquad \qquad \frac{1 - \beta(1 - \delta_m)}{\phi\eta\beta} = \frac{P^*(Y_2^*)^{1 - \eta}(y_d^*)^{\eta} + PY_2^{1 - \eta}(y_f^*)^{\eta}}{M^*}$$

$$wL_{1} = (1 - \theta)\nu Y \qquad wL_{1}^{*} = (1 - \theta)\nu Y^{*}$$

$$wl_{d} = (1 - \theta)(1 - \phi)\eta PY_{2}^{1 - \eta}y_{d}^{\eta} \qquad wl_{f}^{*} = (1 - \theta)(1 - \phi)\eta PY_{2}^{1 - \eta}(y_{f}^{*})^{\eta}$$

$$w^{*}l_{d}^{*} = (1 - \theta)(1 - \phi)\eta P^{*}(Y_{2}^{*})^{1 - \eta}(y_{d}^{*})^{\eta} \qquad w^{*}l_{f} = (1 - \theta)(1 - \phi)\eta P^{*}(Y_{2}^{*})^{1 - \eta}y_{f}^{\eta}$$

$$L = l_{f}^{*} + l_{d} + L_{1} \qquad L^{*} = L_{1}^{*} + l_{d}^{*} + l_{f}$$

$$K = k_{f}^{*} + k_{d} + K_{1} \qquad K^{*} = K_{1}^{*} + k_{d}^{*} + k_{f}$$

$$Y = (K_{1}^{\theta}L_{1}^{1 - \theta})^{\nu}Y_{2}^{1 - \nu} \qquad Y^{*} = ((K_{1}^{*})^{\theta}(L_{1}^{*})^{1 - \theta})^{\nu}(Y_{2}^{*})^{1 - \nu}$$

$$P = (1 - \nu)Y \qquad P^{*} = (1 - \nu)Y^{*}$$

 $c + c^* + \delta(K + K^*) + \delta_m(M + M^*) = Y + Y^*$

Note that when $\tau = 1$ both countries are identical, therefore all home and foreign quantities and prices are identical, so I will omit for now the asterisk. One can show that $(k_d + k_f)/K_1 = (l_d + l_f)/L_1 = \xi$, with $\xi = (1 - \phi)(1 - \nu)\eta/\nu$. Also note that the parameters of the utility function are set so that L = 1 in steady state, so that we obtain the following system of equation for technology, physical capital and aggregate output:

$$\begin{split} M &= \frac{\phi\beta(1-\nu)\eta}{1-\beta(1-\delta_m)}Y\\ K &= \frac{(1-\phi)\theta\beta(1-\nu)\eta}{1-\beta(1-\delta)}Y\\ Y &= 2^{1-\nu}\left(\frac{1}{1+\xi}\right)^{\nu}\left(\frac{\xi}{1+\xi}\right)^{(1-\phi)(1-\nu)}K^{\theta(\nu+(1-\nu)(1-\phi))} \end{split}$$

We can combine these equations and obtain the steady state values for technology and physical capital

$$M = \zeta_1^{\frac{\theta(\nu+(1-\nu)(1-\phi)}{\zeta_3}} \zeta_2^{\frac{1}{\zeta_3}}$$
$$K = \zeta_1 M$$

with

$$\begin{aligned} \zeta_1 &\equiv \frac{\theta(\nu + (1 - \nu)(1 - \phi)\eta)}{\phi(1 - \nu)\eta} \frac{1 - \beta(1 - \delta_m)}{1 - \beta(1 - \delta)} \\ \zeta_2 &\equiv \frac{\phi\beta(1 - \nu)\eta}{1 - \beta(1 - \delta_m)} 2^{1 - \nu} \left(\frac{1}{1 + \xi}\right)^{\nu} \left(\frac{\xi}{1 + \xi}\right)^{(1 - \phi)(1 - \nu)} \\ \zeta_3 &\equiv 1 - \phi(1 - \nu) - \theta(\nu + (1 - \phi)(1 - \nu)). \end{aligned}$$

All the other quantities follow by plugging in these values in the respective equations. Because financial market structure only in the economy with uncertainty, the deterministic steady state is the same whether or not financial markets are complete.

C.3 The incomplete markets model with country-specific shocks only

This appendix presents - for completeness - the results on the model version with incomplete financial markets and country-specific productivity shocks only. The purpose is to show that the main results on investment and GDP synchronization remain unaltered, even when abstracting from multinational-specific shocks under incomplete markets. Figure 5 shows a similar plot with respect to investment synchronization as in the main text. In panel a), I show the benchmark results on the complete markets model with both shocks. Panel b) is new and shows the incomplete markets model with productivity shocks only, panel c) and d) showing investment synchronization and FDI positions as functions of FDI openness parameter τ for the incomplete markets economy with both shocks. It is evident from panel b) that the presence of technology capital only leads to the prediction of increased investment synchronization, even when markets are incomplete and we abstract from shocks to multinational activity. The mechanisms behind this finding are exactly the same as outlined in the main text for the complete financial markets economies: When countries are more open to FDI, the returns of technology capital increase for both domestic and foreign multinationals, hence investment in technology capital increases in both countries. Because returns on all capital types are equalized within firms, multinationals increase investment in physical capital both at home and abroad, leading to an increase in investment co-movement.

Regarding GDP co-movement, figure 6 shows that the findings in the incomplete market model with productivity shocks only are very similar to the complete market model with productivity only.



Figure 5: FDI openness and Investment synchronization, details on the incomplete markets model



Figure 6: FDI openness and GDP synchronization, details on the incomplete markets model